Author's response to reviews

Title: Construct validity of functional capacity tests in healthy workers

Authors:

Sandra E Lakke (a.e.jorna-lakke@pl.hanze.nl)
Remko Soer (R.Soer@cvr.umcg.nl)
Jan HB Geertzen (j.h.b.geertzen@umcg.nl)
Harriët Wittink (harriet.wittink@hu.nl)
Rob KW Douma (k.w.douma@pl.hanze.nl)
Cees P Van der Schans (c.p.van.der.schans@pl.hanze.nl)
Michiel F Reneman (m.f.reneman@cvr.umcg.nl)

Version: 5 Date: 22 May 2013

Author's response to reviews: see over
To: Dr J.B. Staal, Section Editor of BMC Musculoskeletal Disorders  
Section: Rehabilitation, physical therapy and occupational health  
From: Sandra Lakke, on behalf of the co-authors  
Date: May 22, 2013  
Re: Accepted manuscript: Construct validity of functional capacity tests in healthy workers

Dear Dr. Staal,

Thank you for reviewing and acceptance of our manuscript. We have made changes according to Editorial Requirements.

- We added an abbreviations section
- We added a competing interests section

These two changes in the manuscript are indicated using red type.

Sincerely,

On behalf of all authors,

Sandra Lakke

E-mail: a.e.jorna-lakke@pl.hanze.nl